Author’s response to reviews

Title: Eplerenone attenuates pathological pulmonary vascular rather than right ventricular remodeling in pulmonary arterial hypertension

Authors:

Mario Boehm (mario.boehm@innere.med.uni-giessen.de)
Nadine Arnold (n.d.arnold@sheffield.ac.uk)
Adam Braithwaite (a.braithwaite@sheffield.ac.uk)
Josephine Pickworth (j.a.pickworth@sheffield.ac.uk)
Changwu Lu (Changwu.Lu@innere.med.uni-giessen.de)
Tatyana Novoyatleva (Tatyana.Novoyatleva@innere.med.uni-giessen.de)
David Kiely (david.kiely@sth.nhs.uk)
Friedrich Grimminger (Friedrich.Grimminger@innere.med.uni-giessen.de)
Hossein Ghofrani (ardeschir.ghofrani@innere.med.uni-giessen.de)
Norbert Weissmann (Norbert.Weissmann@innere.med.uni-giessen.de)
Werner Seeger (Werner.Seeger@innere.med.uni-giessen.de)
Allan Lawrie (A.Lawrie@sheffield.ac.uk)
Ralph Schermuly (Ralph.Schermuly@innere.med.uni-giessen.de)
Baktybek Kojonazarov (Baktybek.Kojonazarov@innere.med.uni-giessen.de)

Version: 3 Date: 24 Jan 2018

Author’s response to reviews:

Point-by-point response letter

Q1: Thank you for re submitting the Authorship Change form. We have noted that, at least some of the signatures, are copy-paste signatures from other documents. Please, provide the original signatures and scan the documents for us.
R1: The original signatures and scanned documents were sent to BMCSeriesEditorial@biomedcentral.com. All authors agree with the inclusion of Dr. Kiely to the author list. Please let me know if I can be of further help on this issue.

Q2: We note that, in Ethics approval and consent to participate, you state: 'The local authorities for animal research approved the study protocol (Regierungspräsidium Giessen, Germany, Gi 32/2013 and The UK Home Office under PPL 40/3517)'. Please, could you clarify if the approval was provided by the ethics committees of these authorities?

R1: We added the following statement to the declarations section:

All experiments were performed according to the institutional guidelines that comply with national and international regulations (EU directive 2010/63). The local authorities for animal research approved the study protocol (Regierungspräsidium Giessen, Germany, Gi 32/2013 and The UK Home Office under PPL 40/3517).